Author’s response to reviews

Title: HIV Testing Awareness and Service Uptake Among Female Heads of Household in Rural Mozambique: Results from a Province-wide Survey

Authors:

Heather N Paulin (hnpaulin@gmail.com)
Meridith Blevins (meridith.blevins@Vanderbilt.Edu)
John R Koethe (john.r.koethe@Vanderbilt.Edu)
Nicole Hinton (nicole.l.hinton@Vanderbilt.Edu)
Lara M. E. Vaz (lmevaz@gmail.com)
Alfredo Vergara (nicole.l.hinton@Vanderbilt.Edu)
Abraham Mukolo (abraham.mukolo@Vanderbilt.Edu)
Elisee Ndatimana (elisee.ndatimana@vanderbilt.edu)
Troy Moon (troy.moon@Vanderbilt.Edu)
Sten H Vermund (sten.vermund@Vanderbilt.Edu)
C. William Wester (william.wester@Vanderbilt.Edu)

Version: 2 Date: 30 June 2014

Author’s response to reviews:

Requested changes were made on June 30th:

1. Line numbering was inserted
2. Competing interest clause was inserted
3. Author email addresses were inserted